Author's response to reviews

Title: Chronic and acute effects of walnuts on antioxidant capacity and nutritional status in humans: a randomized, cross-over pilot study

Authors:

Diane L. McKay (diane.mckay@tufts.edu)
C-Y. Oliver Chen (oliver.chen@tufts.edu)
Kyung-Jin Yeum (kyungjin.yeum@tufts.edu)
Nirupa R. Matthan (nirupa.matthan@tufts.edu)
Alice H. Lichtenstein (alice.lichtenstein@tufts.edu)
Jeffrey B. Blumberg (jeffrey.blumberg@tufts.edu)

Version: 6 Date: 30 April 2010

Author's response to reviews:

Minor typographical errors were corrected.